Author's response to reviews

Title: Acceptability and impact on anthropometry of commercial available versus locally developed Ready-to-Use Therapeutic Food in pre-school children in Vietnam.

Authors:
Tran T Nga (thuynga1997@gmail.com)
Marie Nguyen (marienguyen21@yahoo.fr)
Roger Mathisen (rmathisen@unicef.org)
Do TB Hoa (trankexuong@yahoo.com)
Nguyen H Minh (nguyenhongminh3215@yahoo.com.vn)
Jacques Berger (jacques.berger@ird.fr)
Frank T Wieringa (franck.wieringa@ird.fr)

Version: 4 Date: 7 July 2013

Author's response to reviews:

Dear Editor,

Please find attached the re-revised manuscript on the acceptability and impact on anthropometry of commercial available versus locally developed Ready-to-Use Therapeutic Food in pre-school children in Vietnam.

I have highlighted in yellow the few last remaining issues of Reviewer 1 and the comment of the Editor in the discussion. Specifically, we now also report on the weight gain in the children over the 2 week intervention for each product (Plumpy’Nut vs HEBI). As height was only measured at baseline and after 1 month, we do not have data on height increments over the 2 week intervention periods for both products. In the abstract, we have changed +/- to SD as requested, and added data on the acceptability of both products.

We hope the manuscript is now acceptable for publication. If more questions remain, please do not hesitate to contact me.

Best wishes

Frank Wieringa
Senior Researcher
UMR-204 NutriPass - Prevention of Malnutrition
IRD-Centre de Montpellier, France